Author's response to reviews

Title: Involving deprived communities in improving the quality of primary care services: does participatory action research work?

Authors:

Peter G Cawston (p.cawston@nhs.net)
Stewart W Mercer (sm83z@clinmed.gla.ac.uk)
Rosaline S Barbour (r.barbour@dundee.ac.uk)

Version: 4 Date: 7 June 2007

Author's response to reviews:

We have made the formatting changes requested;
Country of Origin added for each author
Removal of word counts, key words, running title
Introduction changed to Background
Methodology changed to methods
Guarantor and contributorship moved to acknowledgements, with funding and ethical approval.
Authors contributorship now includes indication that all authors approved final version
Fig titles moved to text after references.
Figures cropped as close as possible.